Author's response to reviews

Title: Health care seeking behavior for diarrhea in children under 5 in rural Niger: results of a cross-sectional survey

Authors:

Anne-Laure Page (anne-laure.page@epicentre.msf.org)
Sarah Hustache (sarah.hustache@yahoo.fr)
Francisco J Luquero (francisco.luquero@epicentre.msf.org)
Ali Djibo (ali_djibo@yahoo.fr)
Mahamane Laouali Manzo (manzolic@yahoo.fr)
Rebecca F Grais (rebecca.grais@epicentre.msf.org)

Version: 4 Date: 19 May 2011

Author's response to reviews: see over
May 19th, 2011

Dr Margaret Elizabeth Kruk
BMC

Reference: Submission of revised manuscript 1637854116487989

Dear Dr. Kruk,

Thank you for accepting our manuscript, “Health care seeking behavior for diarrhea in children under 5 in rural Niger: results of a cross-sectional survey.”

We have made the requested changes to the format of the manuscript. The abstract was updated in the submission system. The title page and citations were modified to fit your format. Finally, the figure was modified and cropped to the size. We hope the manuscript is now in an acceptable format.

Sincerely,

Anne-Laure Page
Epicentre
8 rue Saint Sabin
75011 Paris France